Author’s response to reviews

Title: MANAGEMENT OF SEXUAL PARTNERS OF PREGNANT WOMEN WITH SYPHILIS IN NORTHEASTERN BRAZIL - A QUALITATIVE STUDY

Authors:

Ana Rocha (ana_lumen@hotmail.com)
Maria Araujo (ali.araujo@hotmail.com)
Angélica Miranda (espinosa@ndi.ufes.br)
Rodolfo de Leon (email.gomezr@paho.org)
Geraldo da Silva Junior (geraldobezerrajr@yahoo.com.br)
Lea Vasconcelos (leadpg@ig.com.br)

Version: 4 Date: 07 Jan 2019

Author’s response to reviews:

Dear Editor and Reviewers of the BMC Health Services Research Journal,

We are resubmitting the manuscript entitled “MANAGEMENT OF SEXUAL PARTNERS OF PREGNANT WOMEN WITH SYPHILIS IN NORTHEASTERN BRAZIL - A QUALITATIVE STUDY” (BHSR-D-18-00060).

We appreciate the reviewers’ contributions and would like to point out that the suggested recommendations were promptly addressed.

1. Please change the title of your ‘Introduction’ to ‘Background’. – Done. Line 60

2. Please represent authors' names using their full initials, not their full/sur- name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors. – Done. Line 538

3. Currently, the contributions of authors Silva Junior GB do not automatically qualify them for authorship. Please provide clarification on their contributions, or remove their names
from the list of authors and place them in the “Acknowledgements” section instead. Please note that any changes in the list of authors requires the completion of the “Change in authorship” form. – GBSJ is author and contributed considerably with the article. The authors' contributions were better detailed. – Lines 538 – 543.

4. Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them. This includes simple titles like 'Figure 1'. – Done.

5. Please provide figure titles and legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them. – Done. Line 660.

6. Please ensure that all figures/tables and supplementary files are cited within the main text (this does not include the declarations). Any items which are not cited may be deleted by our production department upon publication. Currently, figure 1 is not cited. – Done. The figure was cited. – Lines 402-403.

7. Please remove the response to reviewers document from the end of your manuscript file as this is no longer needed in the publication process. - OK

8. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. - OK